Author's response to reviews

Title: Time trends in social differences in nutrition habits of Lithuanian population: 1994-2010

Authors:

Vilma Kriaucioniene (vilmabor@yahoo.co.uk)
Jurate Klumbiene (jurklumb@vector.kmu.lt)
Janina Petkeviciene (janpet@vector.kmu.lt)
Edita Sakyte (etykas@yahoo.com)

Version: 2 Date: 21 October 2011

Author's response to reviews: see over
Dear Editors,

We would like to respond to your message:

‘Thank you for your recent submission of the manuscript detailed above. Before we can proceed with peer review we will need you to make some changes to the manuscript. We strongly urge you to make these changes promptly, as we cannot start the peer review process until we have received a version containing the changes.

Further Requirements:
1. Ethics - Please document within the methods section of your manuscript the specific name of the organization that granted ethical approval to your study going ahead.’

We included in the Methods section this sentence:
‘The Lithuanian Bioethics Committee approved all surveys.’

Respectfully,

On behalf of all authors
Vilma Kriaucionienė